Author's response to reviews

Title: Incidence and pattern of injuries among residents of a rural area in South-Western Nigeria: a community-based study.

Authors:

Omoniyi A Olawale (omoniyi_olawale@yahoo.com)
Eme T Owoaje (emeowoaje@yahoo.com)

Version: 7 Date: 6 September 2007

Author's response to reviews: see over
Title: Incidence and pattern of injuries among residents of a rural area in Southwestern Nigeria: a community-based study (MS 5190095651201410)

Authors:

Olawale Omoniyi (omoniyi_olawale@yahoo.com)

Eme Owoaje (emeowoaje@yahoo.com)

Version: 4 Date: 6 September, 2007

Cover letter: see over
September 6, 2007

The Editor

BMC Public Health

Re: Incidence and pattern of injuries among residents of a rural area in South western Nigeria: a community-based study (MS 5190095651201410)

Thank you for the email.

We thank the editors for accepting to publish this work. We are sending a re-formatted version of the article.

Yours sincerely,

Olawale Omoniyi

Corresponding author